Author's response to reviews

Title: Pullout strength of pedicle screws with cement augmentation in severe osteoporosis: A comparative study between cannulated screws with cement injection and solid screws with cement pre-filling

Authors:

Lih-Huei Chen (LHCHEN2132@adm.cgmh.org.tw)
Ching-Lung Tai (taicl@mail.cgu.edu.tw)
De-Mei Lee (dmlee@mail.cgu.edu.tw)
Po-Liang Lai (polianglai@gmail.com)
Yen-Chen Lee (m9825006@mail.cgu.edu.tw)
Chi-Chien Niu (niuchen@adm.cgmh.org.tw)
Wen-Jer Chen (chenwenj@adm.cgmh.org.tw)

Version: 7 Date: 29 January 2011

Author's response to reviews: see over
Jan. 29, 2011

Manuscript ID: 1338674989411171

Title: Pullout strength of pedicle screws with cement augmentation in severe osteoporosis: A comparative study between cannulated screws with cement injection and solid screws with cement pre-filling

Dear Editor:
We are pleased to receive your notification that our manuscript has been accepted for publication. We have checked our manuscript carefully for any typing or content errors, and the following formatting changes have been made.
1. All three tables have been created using a Tables tool.
2. All highlighting and tracking in previous manuscript have been removed.
3. The section of "Authors' contributions" has been formatted into a paragraph.

Sincerely,

Ching-Lung Tai, PhD  (Corresponding author)
Graduate Institute of Medical Mechatronics
Department of Mechanical Engineering
Chang Gung University
259, Wen-Hua 1st RD., 333,
Kweishan, Taoyuan, Taiwan, R.O.C.
Tel: +886-3-2118800 ext. 5342
Fax: +886-3-2118050
E-mail:taicl@mail.cgu.edu.tw